Joint Field Office Situation Report 
Human Services

Date: 10/5/05

1603:
	Registrations
	1,018,560 (COB 10/3)

	Total Payments
	$2,099,348,576.25

	Designated Parishes 
	31


1607:

	Registrations
	248,538 (COB 10/3)

	Total Payments
	$232,751,178.42

	Designated Parishes 
	21


	DRCs 

	Fixed
	27

	Mobile
	2


	Section Reporting: 
	HS IA Other Programs/Special Needs Group

	Group Leader Contact Info:
	Margaret Braun 225-346-4157; 703-547-7668

	1.
	SITUATION:

	
	A.
	Action Items Completed 
	1603/1607: 

Completed “Special Needs” Resources Guide and distributed to SN Group.  



	
	B.
	Action Items for next Operational period
	1603/1607: 

Identify date, staff and time for Special Needs Housing meeting.   

Identified staff  for the Special Needs Housing meeting: IA Housing,  Transitional Team, IAO, Special Needs; U.S. Dept. of Health and Human Services; State IAO, and  Dept Social Services (Housing); ERO; ESF 6 & 8; DHS Attorney, with Office for Civil Rights and Civil Liberties; meeting date and time TBD.

Meet with Transitional Housing Team, Groom Road contractor, and IAO to discuss elements of Special Needs Housing meeting, current status of site operations, and status of available ADA units.  

	
	C.
	General Status of Operations
	SN Group Lead continues coordination with JFO staff to address the need for clear strategy and plan for implementation of  housing Special Needs applicants. 
Hiring process for 3 new DAEs is ongoing. Staff may be deployed by the end of the month. 

1603:
Special Needs (SN): Referrals Cumulative: 345; Pending:24.
Crisis Counseling (CC): Funds have been obligated for the ISP undeclared parishes.  Initial ISP for declared parishes should be received today. The recommendation will not be for the $10,490,778.00 requested, but will be for approximately $5,186,692.00.   
Disaster Unemployment Assistance (DUA): FEMA JFO Program representative questioned the State’s workload activity count; appears that regular UI applications may be erroneously included in the count. The USDOL Director shared the same concern, and has deployed a 2-person team to investigate. 

Disaster Legal Services (DLS): Activity Report is pending.
Other Needs Assistance (ONA): Ongoing.
1607:

Special Needs (SN): Referral Cumulative:8; Pending: 0.
Crisis Counseling (CC): Initiation of program pending.
Disaster Unemployment Assistance (DUA): FEMA has not received a request from the state for funding for Hurricane Rita. 
Disaster Legal Services (DLS): Initiation of program pending.
Other Needs Assistance (ONA): Ongoing.

	
	D.
	Staffing Needs
	

	
	
	
	(3) Pending completion of DAE hiring process.
	# of Staff on Location
	# of Staff pending check-in
	# of Staff still needed

	
	
	
	
	9
	(3)
	0

	2.
	CURRENT ISSUES OR PROBLEMS: 
1603/1607: NA


	Section Reporting: 
	ESF-11

	Group Leader Contact Info:
	Mike McIlwain

	1.
	SITUATION:

	
	A.
	Action Items Completed 
	1603: 

Disaster Mass Care Feeding (FNS)  

Food: 
State DA Orders Since Last Report
Nothing new reported from the previous report

State DA Shipments to Disaster Organizations Since Last Report
             Nothing new reported from the previous report

In response to Hurricane Katrina, Louisiana has issued 329,993 EBT cards to victims for an issuance of $191,998,414, which includes $73,652,560 for October benefits automatically to those who received disaster benefits in September.  
FNS regional Food Stamp Program staff are participating in a Jefferson Parish disaster recovery event sponsored by the National Alliance of Faith and Justice and the Cops and Clergy Initiative.  Over 1,000 are expected to attend.

1607:

In response to Hurricane Rita, Louisiana has issued 73,576 EBT cards to victims for an issuance of $32,030,999.

On Tuesday October 4, regional teams visited 20 food stamp sites, shelters and WIC program sites in Louisiana. No significant food stamp issues were identified

WIC:  Supplemental Food Program Division (SFPD) has given permission for other state agencies to accept Louisiana October WIC vouchers because of Hurricane Rita.  

	
	B.
	Action Items for next Operational period
	1603: Continue Disaster Food Stamp Program for affected parishes

1607: Continue Disaster Food Stamp Program for affected parishes

	
	C.
	General Status of Operations
	1603: Ongoing

1607: Ongoing

	
	D.
	Staffing Needs
	

	
	
	
	
	# of Staff on Location
	# of Staff pending check-in
	# of Staff still needed

	
	
	
	
	3
	1
	0

	2.
	CURRENT ISSUES OR PROBLEMS: 

1603: Nothing significant reported October 4, 2005

1607: Nothing significant reported October 4, 2005

	
	


	Section Reporting: 
	Housing

	Group Leader Contact Info:
	Juan Gil/Maggie D.

	1.
	SITUATION:

	
	A.
	General Status of Operations
	A reporting meeting was held on 10/4/05 and 10/5/05.  

· Reports will be reconciled so that all #s are consistent.

· An interim process was established to provide Dewberry w/occupancy numbers that will be entered into HAC database for consistent reporting.  When contractors begin tracking occupancy agreements, contractors will update the HAC database with those figures.  

· Existing commercial sites continue to be identified.

	
	B.
	Major Priorities Accomplished
	RFO’s today –Slidell I-59 MHP/St. Tammany Parish (20)

                       Ken Albin’s/Tangipahoa Parish (2)

	
	C.
	Staffing Needs
	

	
	
	
	
	# of Staff on Location
	# of Staff en route
	# of Staff still needed

	
	
	
	
	
	0
	20

	2.
	CURRENT ISSUES OR PROBLEMS:  


	Section Reporting: 
	Registration Sweep Teams

	Group Leader Contact Info:
	Pat Lord

	1.
	SITUATION:

	
	A.
	Action Items Completed 
	1603: 

· We have completed all of the major shelters for registration intake.

1607:  

· We have completed over 1000 registrations.

	
	B.
	Action Items for next Operational period
	1603:    Complete special requests for registration intake. 

1607:     Determine what additional shelters we need to visit.

	
	C.
	General Status of Operations
	1603:

· 1SWEEP TEAM IS OUT ON THE FIELD TODAY.

· We have 20 staff members to respond to all requests

· The State is contacting all affected Parishes to see if they need a sweep team.

· We are finding most people are registered and have program questions.

1607:

· 4 SWEEP TEAMS ARE OUT IN THE FIELD TODAY. 

· Checking ARC list to see if any shelters need our help with registrations.

	
	D.
	Staffing Needs
	

	
	
	
	
	26# of Staff on Location
	# of Staff pending check-in
	# of Staff still needed

	
	
	
	
	2 FEMA Staff / 22 contract staff 
	0
	0

	2.
	CURRENT ISSUES OR PROBLEMS: 
1603:
1607:


	Section Reporting:  
	VOLAG

	Group Leader Contact Info: 
	Harry Noftsker

	1.
	SITUATION:

	
	A.
	Action Items Completed 
	1603: 

1607:

	
	B.
	Action Items for next Operational period
	1603:

1607:

	
	C.
	General Status of Operations
	1603: 

CATHOLIC CHARITIES:

-On Tuesday, October 4th Catholic Community Services of Baton Rouge serviced 118 applicants and distributed $60,541 in aid for rent, utilities, transportation, and other needs for displaced persons.

DONATIONS:

Met with State donations, FEMA housing and came up with agreed components of a trailer living kit.  

Plan is to locate warehouse to assemble kits for delivery to GSA @ Choctaw & Sherwood.  Donations group working with NVOAD and FEMA looking to secure components for 70,000-100,000 finished kits.

One group has purchased 150 kitchen, and first aid kits, also will purchase up to 1000 kits.

3000 school kits will be delivered to East Baton Rouge Oct 5.

Hammond –Convoy of Hope warehouse is now using AIDMATRIX.  Looking for warehouse for long term recovery.

SOUTHERN BAPTIST RELIEF:

1. Completed 78,261 meals.  Chain Saw recovery work continuing across the state.   Continuing Blue roof project with support from Adventist Community Services and Corps of Engineers.

2. Feeding to continue with 22 Kitchens working with the Red Cross and The Salvation Army.  Continuing to review request for mudout units in the New Orleans area.

3. Anticipate feeding and recovery operations to continue min. 30 -45 days.

LANO:

LANO is still troubleshooting and coordinating specific needs with local nonprofit agencies in the area.  Our immediate issues are still trying to provide daycare for children from ages 0-5.  We are also working with our members to make sure necessary documents have been submitted with their Requests for Public Assistance applications.

  

ACS:

The New Iberia Multi-Agency Warehouse Reports:

Offloaded 2 trucks, top items

            Bleach

Clothes

Snacks

           (Also received one fork lift)

Shipped out 53 pallets since then, top items

            Bedding

            Cleaning Supplies

Sacks

                      

Donated goods needed: cleaning supplies, bedding

Needs for the warehouse:  1 -2 forklifts (forklifts that we now have are being recalled by the lenders), additional pallet jacks, boxes for packing items for distribution.

UNITED SIKHS:

· Attended the Interfaith Emergency Relief Teams meeting at the Episcopal Church in downtown Baton Rouge.  The meeting discussed whether there was a need of an interfaith group to be involved in the relief process and what the role of such a group would be.

· Went price shopping for pillows and came to a deal with “Big Lots” for a good price for 60 pillows which we may be able to receive tomorrow.

· Delivered approximately 60 blankets to Allen Chapel, of which some were also supplied to a neighboring shelter who have no bedding and currently have only basic supplies.  We also arranged a date when UNITED SIKHS will be providing langar (free vegetarian lunch) for the residents of the shelter, and assessed the needs of the shelter.  Allen Chapel is in need of long-term supplies for the shelter. The building needs renovating to accommodate those in the shelter, currently they house approximately 60 shelters but with other shelters in the area closing, they expect that number to increase over the coming weeks.  People are also being housed in the homes of the shelter workers.

· Further research was conducted to assess the needs of Sikh families after the Hurricanes, and put aside some supplies for those in need.

· Preliminary research was carried out for the UNITED SIKHS working paper on “The Role of Minorities as Service-Providers in Emergency Situations”

GOODWILL:

Goodwill was successful in facilitating the delivery of a hospital bed and hoyer lift for a New Orleans quadriplegic evacuee today, who is being sheltered in Baton Rouge, LA.    

Goodwill is continuing to work on obtaining power wheelchairs for three New Orleans area evacuees in Baton Rouge, Jefferson Parish and Winnsboro, LA.  

Goodwill is continuing to provide $20.00 of merchandise with a Red Cross Shelter voucher at operating Goodwill stores in the Baton Rouge, Denham Springs, Gonzales, Houma and Algiers areas.  

SCIENTOLOGY DISASTER RESPONSE:

Spiritual first aid assists delivered to persons in need 74 cumulative 14,151 

Number of people helped as needed 504 cumulative 75,349

Number of Way to Happiness booklets passed out 326 cumulative 583

We are working on getting the Criminon Program implemented in Jefferson Parish. The ED (Executive Director) Criminon was here and met with the warden in Jefferson parish and they are working on getting the sheriffs deputies trained on Criminon so that is a very exciting cycle but not one we are saying is a done as it is not done.

We send an assessment team to Cameron and they are on their way back today to help.  The other sites mentioned before are still in operation and we are still giving out supplies (water, food, etc as needed).

 

CAN: (COORDINATED ASSISTANCE NETWORK)

Case management is currently available.  Case managers have access to view all clients currently in the system in order to provide more integrated services and referrals.

We are developing processes for input of the resource agencies and services. 

This information will be accessed by case managers to provide client assistance.

To date, CAN has registered 1185 agency users and 82,038 clients.  This provides case history for clients across geographical boundaries as they relocate and/or return home.

ARC:

246,198 meals this period:  6,950,168 meals to date

267,588 snacks this period: 11,170,028 snacks to date                   

98 shelters this period; 256 shelters to date.

15,351 population period; 141,018 populations to date.

21,218 rooms being utilized for transition accommodations in LA.

Community Services

There are 18 kitchens cooking and catering operations LA

Families Re-Connecting

Calls to 1-877-LOVED 1S total 187,070 with 2,118 calls received in the past 24 hours.

Financial Assistance

Financial assistance will no longer be given via debit cards.  Checks, Western Union and electronic fund transfer will still be used to distribute relief funds.

TSA:

Food Served

 

Today*

Running Total

Prepared Meals (Hot or Cold)

56,404

943,913

Sandwiches

 

1,838

137,755

Cold Drinks (Soda/Juice/Water)

70,411

835,183

Snacks (Donuts, Cakes, Chips)

38,321

574,760

Other 

 

0

34,496



	
	D.
	Staffing Needs
	

	
	
	
	
	# of Staff on Location
	# of Staff pending check-in
	# of Staff still needed

	
	
	
	
	6
	7
	0

	2.
	CURRENT ISSUES OR PROBLEMS: 

1603:

1607:


	Section Reporting:  
	U.S. SBA

	Group Leader Contact Info:
	Carl N. Gaspari

	1.
	SITUATION:

	
	A.
	Action Items Completed 
	1603: Customer Service Representatives are assigned to all DRCs to answer questions and assist individuals and business owners complete their SBA Application.  

A Personnelist is now on site to begin hiring local staff.

Identified location for a Business Recovery Center in NO

1607:

	
	B.
	Action Items for next Operational period
	1603: 

Begin training of new CSR staff  for DRCs on 10/5/05

1607:

	
	C.
	General Status of Operations
	1603: 

SBA continues to search out conventional lodging for members or our staff. 

SBA continues search for a location in New Orleans to support a Business Recovery Center. 

Local hiring of CSRs and administrative staff is ongoing. 

1607:

	
	D.
	Staffing Needs
	

	
	
	
	
	# of Staff on Location
	# of Staff pending check-in
	# of Staff still needed

	
	
	
	
	LA 358  TX 70
	2
	142

	2.
	CURRENT ISSUES OR PROBLEMS: 
1603:  Locating lodging to support staff in close proximity to the DRCs and JFO.  

1607:


	Section Reporting: 
	Barbara Copeland/Kent Sheets

	Group Leader Contact Info:
	Inspection Service Coordinators

	1.
	SITUATION:

	
	A.
	Action Items Completed 
	1603: 

There are 1216 inspectors performing inspections. 

1607: 
There are 279 inspectors performing inspections.

	
	B.
	Action Items for next Operational period
	1603:
1607:

	
	C.
	General Status of Operations
	1603:

1607:

	
	D.
	Staffing Needs
	

	
	
	
	
	# of Staff on Location
	# of Staff pending check-in
	# of Staff still needed

	
	
	
	
	2
	0
	0

	2.
	CURRENT ISSUES OR PROBLEMS: 
1603:
1607:


	Section Reporting: 
	NPSC Liaison

	Group Leader Contact Info: 
	Pamela Glasschroeder 703-785-2661

	1.
	SITUATION:

	
	A.
	General Status of Operations
	1603/1607: 

· Reviewing Special Needs, Congressional and escalated cases for varies entities with in the JFO

· Monitoring reports and queues

· Relating information between the NPSCs and JFO 

	
	B.
	Major Priorities Accomplished
	1603/1607:
· Provided answers to questions regarding Auto-dialer

· Assisted special needs applicant with information regarding how to have brothers remains retrieved and removed from damaged property.  Case was turned over to DMORT team which made contact with applicant and provided appropriate referrals.

	
	C.
	Staffing Needs
	

	
	
	
	
	# of Staff on Location
	# of Staff pending check-in
	# of Staff still needed

	
	
	
	
	3
	0
	0

	2.
	CURRENT ISSUES OR PROBLEMS: 


	Section Reporting: 
	DRC OPS

	Group Leader and Contact Info:
	Don Baggett

	1.
	SITUATION:

	
	A.
	General Status of Operations
	DRC at Belle Chasse will be open daily from 8am – 6pm.  All other DRC’s will be open daily from 9am – 7pm.

 Fixed DRC sites now in 27 locations:

Caddo(Shreveport)

Ouacita(Monroe)

Rapides(Alexandria)

Lafayette(Lafayette)

Iberville(Plaquemine)

Plaquemines (Belle Chase)

Webster (Minden)

E. Baton Rouge (Baton Rouge)

Washington(Bogalusa)

Avoyelles(Marksville)

Washington (Franklinton)

St Tammany(Slidell)

Orleans(Algiers)

St James (Vacherie)

LaFourche(Thibodeaux)

St Charles (Boutte)

St John (LaPlace)

Ascension(Gonzales)

Jefferson(Gretna)

Iberia(New Iberia)

LaFourche(Raceland)

Tangipahoa(Amite)

Livingston (Denham Springs)

Terrebonne(Houma)

Point Coupee (New Roads)

St Tammany(Covington)

Vermillion(Abbeville)

MDRC’s:

Cruise Ship at Port of New Orleans

Washington Parish

We have also secured leases on properties at the following locations and will announce openings ASAP(In order of priority):

Tentatively Wednesday 10/8/05 or 10/9/05:

Orleans(New Orleans)

W. Baton Rouge(Port Allen)  

St Helena(Greensburg)  

Jefferson Davis(Jennings)  

Cameron(Grand Lake) ASAP

Calcasieu (Lake Charles) ASAP

St Bernard(Chalmette) ASAP

Concordia(Vidalia) ASAP

Beauregard(DeRidder) ASAP

Lincoln(Ruston) ASAP

Natchitoches(Natchitoches) ASAP

Acadia(Rayne) ASAP

Morehouse(Bastrop) ASAP

St Martin(Breaux Bridge) ASAP

#1 Priority: To secure inspected, approved, and leased sites at the following Parishes:  

Sabine (Many)

St Landry(Opelousas)

Red River(Coushatta)

Franklin(Winnsboro)

LaSalle(Jena)

Caldwell(Columbia)

Tensas(St Joseph)

Jackson(Hodge)

Vernon(Leesville)

E Feliciana(Clinton)

Winn(Winnfield)

Evangeline(Ville Platte)

More locations possibly to follow.

	
	B.
	Major Priorities Accomplished
	

	
	C.
	Staffing Needs
	

	
	
	
	
	# of Staff on Location
	# of Staff en route
 
	# of Staff still needed

	
	
	
	
	488
	?
	600

	2.
	CURRENT ISSUES OR PROBLEMS:   

Approximately 50% of the OFA and Firefighter personnel currently deployed to DRC’s will be reaching the end of their tour of duty by 10/07/05 and will have to be replaced.  We are using various resources to address this problem, including additional hiring under the IA TAC, and working with Admin to obtain Local Hires.

DFTO & IA Trainers are now on-site at the JFO to train all DRC personnel.


	Section Reporting: 
	ESF-6 Mass Care

	Group Leader Contact Info:
	Roger Ince; Desk: 225-334-7758; Cell: 916-541-0683

	1.
	SITUATION:

	
	A.
	Action Items Completed 
	1603: 

At current time it is not possible to separate Katrina/Rita items; working on that and hope to have it available for the next reporting period.

 1. ARF: Submitting ARF for the American Red Cross to re-supply multiple kitchens.

2. Miscellaneous: Staffing has been contacted and one person is awaiting confirmation for replacement of Baton Rouge staff leaving (Johannes Troost left 10/5/05). Staff for Lake Charles AFO arriving 10/7/05 (Steve Schroder). Staff for New Orleans JFO arriving 10/5/05 (Constantine Cheremeteff).

1607:

	
	B.
	Action Items for next Operational period
	1603: At current time it is not possible to separate Katrina/Rita items; working on that and hope to have it available for the next reporting period.

1. Trying to get maps ordered up to display ARC/SBC and TSA feeding sites throughout the state, updated DRC/shelters throughout the state.

1607:

	
	C.
	General Status of Operations
	1603: At current time it is not possible to separate Katrina/Rita items; working on that and hope to have it available for the next reporting period.

Operations are ongoing—shelters and feeding operations are normal, no problems.

1607:

	
	D.
	Staffing Needs
	

	
	
	
	
	# of Staff on Location
	# of Staff pending check-in
	# of Staff still needed

	
	
	
	
	3 at JFO
	2 (replacement staff)
	1

	2.
	CURRENT ISSUES OR PROBLEMS: 
1603: N/A

1607:N/A
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1

